Author's response to reviews

Title: Sagittal plane gait characteristics in subjects with hip osteoarthritis with mild to moderate symptoms compared to healthy controls: A cross-sectional study

Authors:

Ingrid Eitzen (ingrid.eitzen@nimi.no)
Fernandes Linda (linda.fernandes1@gmail.com)
Lars Nordsletten (lars.nordsletten@medisin.uio.no)
May Arna Risberg (mayarna.risberg@nimi.no)

Version: 4 Date: 30 November 2012

Author's response to reviews:

The associate Editor had the following comment:

"There still appears to be an error in Table 2 with the Confidence Intervals around the change in knee joint angle- should the lower bound be -2.43 (ie negative rather than positive)?"

This is of course correct. We apologize for not detecting this error when proofreading. The value is changed to a negative value in the revised upload of Table 2.